Author’s response to reviews

Title: A rare case of synchronous double malignancy of carcinoma breast and esophagus treated successfully: a case report

Authors:

Abhishek Singh (hivneg@gmail.com)
Ishwar Chandra Khare (s_nishant_2050@yahoo.com)
Awadhesh Kumar Dixit (dixitak@yahoo.com)
Kailash Chandra Pandey (drpandey@in.com)
Deepak Kumar Mittal (deepakgray@yahoo.co.in)
Parul Singh (parulophtha@gmail.com)

Version: 9 Date: 16 February 2010

Major revisions (We require the author to make these changes)

Ethnicity: Please include the ethnic origin and nationality of the patient in the case presentation section of the manuscript.
• Ethnic origin and nationality of patient included in case presentation.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
• Manuscript checked.